Author’s response to reviews

Title: Progression of planovalgus deformity with age in patients with cerebral palsy

Authors:

Jae Jung Min (osminjaejung@gmail.com)
Soon-Sun Kwon (qrio1010@gmail.com)
Ki Hyuk Sung (skh1219@naver.com)
Kyoung Min Lee (oasis100@empal.com)
Chin Youb Chung (chin@snu.ac.kr)
Moon Seok Park (pmsmed@gmail.com)

Version: 2 Date: 18 Feb 2020

Author’s response to reviews:

Editor Comments:

1. Please include three to ten keywords representing the main content of the article, below your abstract and above the main text of your manuscript.

We have included the following keywords representing our paper as the following:
Pes planovalgus; talo-first metatarsal angle; talo-second metatarsal angle; progression; cerebral palsy

2. As many abbreviations are used throughout the paper, we ask that you include a list of abbreviations (before the declarations section) for reference and ease of reading. All abbreviations should still be defined in the text at first use.

CP: cerebral palsy
PV: pes planovalgus
HV: hallux valgus

Talo-1st MT, talo-2nd MT angle: Talus-first metatarsal angle, talus-second metatarsal angle

GMFCS: Gross Motor and Functional Classification System
AP: Anteroposterior

TAL: Tendoachcilles lengthening

CI: Confidence interval

3. In the “Funding” section of your declarations, please clarify the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

We have made relevant changes to our manuscript.

4. The Author contribution section includes reference to CYJ as an author, who is not listed as an author elsewhere. Please clarify whether CYJ should be included as an author for this manuscript.

If you choose to change your author list you will need to fill out a change in authorship form and send it by email to the Editorial office to be approved by the Editor. The form can be found here: https://www.biomedcentral.com/getpublished/editorial-policies#authorship

We have mistakenly spelled CYC as CYJ. We have made relevant corrections to our manuscript.

Further, the individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section. Currently the contributions of Chin Youb Chung are not described. Guidance and criteria for authorship can be found here:

http://www.biomedcentral.com/submissions/editorial-policies#authorship

We have mistakenly spelled CYC as CYJ. We have made relevant corrections to our manuscript.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

We have uploaded the new manuscript